\documentclass[a4paper,12pt]{article}
%%%%%%%%%%%
%% PACKAGE USE %%
%%%%%%%%%%%
\usepackage{amsfonts}
\usepackage{amsmath}
\usepackage{amsthm}
\usepackage{amssymb}
\usepackage{xcolor}
\usepackage{color, colortbl}
\usepackage{graphicx}
\usepackage{setspace}
\usepackage{booktabs}
\usepackage{mathtools}
\usepackage[longnamesfirst]{natbib} % [round]{natbib}, [longnamesfirst]{natbib}
%\usepackage[font=footnotesize]{caption}
\usepackage[toc,page]{appendix}
\usepackage{hyperref}
\hypersetup{
	colorlinks,
	linkcolor={red!50!black},
	citecolor={blue!50!black},
	urlcolor={blue!95!black}
}
\usepackage{rotating}
\usepackage{siunitx}
\usepackage{epstopdf}
\usepackage{caption}
\captionsetup{labelsep=newline,justification=centering,labelfont={bf,small},textfont={bf,large}} % see caption package documentation
\sisetup{group-separator = {,}}
\usepackage{amstext} % for \text macro
\usepackage{array}   % for \newcolumntype macro
\newcolumntype{L}{>{$}l<{$}} % math-mode version of "l" column type
\newcolumntype{C}{>{$}c<{$}} % math-mode version of "c" column type
\usepackage{multirow}
\usepackage{enumitem} % options for itemize

%\usepackage{amsfonts,graphics,colortbl,color,dsfont,mathtools,geometry,tikz}
%%%%%%%%%%%%
%% ENVIRONMENTS %%
%%%%%%%%%%%%
\newtheorem{thm}{Theorem}
\newtheorem{lemma}{Lemma}
\newtheorem{ill}{Illustration}
\newtheorem{prop}{Proposition}
\newtheorem{cor}{Corollary}
\newtheorem{ass}{Assumption}
\newtheorem{notation}{Notation}
\newtheorem{rem}{Remark}
\newtheorem{ex}{Example}

%%%%%%%%%%%%%%%%%
%% EQUATION RELATED DEFS  %%
%%%%%%%%%%%%%%%%%
\renewcommand{\labelenumi}{(\roman{enumi})}
\newcommand{\bq}[1]{\begin{equation}\label{#1}}
\newcommand{\eq}{\end{equation}}
\newcommand{\bqn}{\begin{eqnarray}}
\newcommand{\eqn}{\end{eqnarray}}
\newcommand{\bqns}{\begin{eqnarray*}}
\newcommand{\eqns}{\end{eqnarray*}}
\newcommand{\non}{\nonumber}
\newcommand{\minus}{\scalebox{0.75}[1.0]{$-$}}

%%%%%%%%%%%%%%%%%%%%
%% CONCEPTS AND ABBREVIATIONS  %%
%%%%%%%%%%%%%%%%%%%%
\newcommand{\func}[1]{\textrm{#1}}
\newcommand{\loge}{\rule{2em}{0.03pt}}
\newcommand{\NID}{\textrm{\rm NID}}
%\newcommand{\se}[1]{\footnotesize\textit{{(#1)}}}
\newcommand{\se}[1]{\footnotesize{(#1)}}
\newcommand{\rutgersays}[1]{\textcolor{blue}{\textbf{RUTGER SAYS: ``#1''}}}
\newcommand{\cc}{\cellcolor{blue!25}}

%%%%%%%%%%%%%%%%
%% LETTER ABBREVIATIONS  %%
%%%%%%%%%%%%%%%%
\newcommand{\mT}{\boldsymbol{T}}
\newcommand{\Ee}{\mathbb{E}}
\newcommand{\unitmat}{{\rm I}}
\newcommand{\dd}{{\rm d}}
\newcommand{\E}{\Ee}
\newcommand{\R}{\mathbb{R}}
\newcommand{\V}{\mathbb{V}\textrm{\rm ar}}
\newcommand{\Z}{\mathbb{Z}}
\newcommand{\N}{\mathbb{N}}
\DeclareMathOperator{\Ran}{Ran}

%%%%%%%%%%%%%%%%%%%
%% GREEK LETTER ABBREVIATIONS %%
%%%%%%%%%%%%%%%%%%%
%\newcommand{\info}{\mathcal{F_{\textit{t}-\text{1}}}}
\newcommand{\info}{\mathcal{F}}
\newcommand{\infoset}[1]{\info_{#1}}
\newcommand{\fisher}{\mathcal{I}_{t|t-1}}
\newcommand{\jacob}{\mathcal{J}_{t|t-1}}
\newcommand{\dphi}{\phi}

%%%%%%%%%%%%%%%
%% LAYOUT DEFINITIONS  %%
%%%%%%%%%%%%%%%

\setstretch{1.5}
\setlength{\textheight}{235mm}
\setlength{\textwidth}{165mm}
\setlength{\oddsidemargin}{-5mm}
\setlength{\evensidemargin}{-5mm}
\setlength{\topmargin}{-1.2cm}
\newcounter{Lcount}

%%%%%%%%%%%%%%%
%% CITATION ALIASES  %%
%%%%%%%%%%%%%%%

%\defcitealias{Goddard(04)}{Goddard et al. (2004)}

%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%
%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%

% to be able to place a figure at the top of the page even when the rest of the page is blank
\makeatletter
\setlength{\@fptop}{0pt}
\makeatother

\begin{document}
\begin{thebibliography}{}

\bibitem[\protect\citeauthoryear{Barndorff-Nielsen, Hansen, Lunde, and
  Shephard}{Barndorff-Nielsen et~al.}{2008}]{Barndorff(08)}
Barndorff-Nielsen, O.~E., P.~R. Hansen, A.~Lunde, and N.~Shephard (2008).
\newblock Realised kernels in practice: Trades and quotes.
\newblock {\em Econometrics Journal\/}~{\em 4}, 1--33.

\bibitem[\protect\citeauthoryear{Brownlees and Gallo}{Brownlees and
  Gallo}{2006}]{Brownlees(06)}
Brownlees, C.~T. and G.~M. Gallo (2006).
\newblock Financial econometric analysis at ultra-high frequency: Data handling
  concerns.
\newblock {\em Computational Statistics \& Data Analysis\/}~{\em 51},
  2232--2245.

\bibitem[\protect\citeauthoryear{Falkenberry}{Falkenberry}{2002}]{Falkenberry(02)}
Falkenberry, T.~N. (2002).
\newblock High frequency data filtering.
\newblock Technical report, Tick Data.

\bibitem[\protect\citeauthoryear{O{'}Hara, Yao, and Ye}{O{'}Hara
  et~al.}{2014}]{OHara(14)}
O{'}Hara, M., C.~Yao, and M.~Ye (2014).
\newblock What's not there: Odd lots and market data.
\newblock {\em Journal of Finance\/}~{\em 69(5)}, 2199--2236.

\end{thebibliography}



\setstretch{1.4}

\noindent \large{Siem Jan Koopman, Rutger Lit, Andr{\'e} Lucas and
 Anne Opschoor, \\ ``Dynamic Discrete Copula Models for High Frequency
 Stock Price Changes'', \emph{Journal of Applied Econometrics}, 2018, 33,
 7, 966--985. }

\subsection*{Data source and preparation}
In this document we provide the steps we took to obtain and prepare the data.
We use data from the Trades and Quotes (TAQ) database of the New York Stock Exchange at a one-second frequency.
VU University Amsterdam has a subscription to the Wharton Research Data Services (WRDS) which offers among other data sources, access to the TAQ database.
It contains intraday transactions data (trades and quotes) for all securities listed on the New York Stock Exchange (NYSE) and American Stock Exchange (AMEX), as well as Nasdaq National Market System (NMS) and SmallCap issues.
We have selected 10 companies from different industries and with different trade intensities for all trading days in 2012. The following corresponding Ticker symbols are USB,
AIG, COF, MS, AXP, BAC, GS, WFC, C and JPM. We analyze the tick-by-tick data without the ``odd-lots'' that represent trades with volumes less than 100 and that are not recorded on the consolidated tape; see for example \citet{OHara(14)}. The data require standard pre-processing. For a review of high-frequency data cleaning procedures; see for example \citet{Falkenberry(02)}.
We apply the cleaning algorithm of \citet{Brownlees(06)} after applying a rudimentary filter corresponding to the cleaning steps P1, P2, P3 and T1, T2, T3 of \citet[p.~8]{Barndorff(08)}.
In cleaning step T2, only trades with the sale condition \{blank, @, *, E, F\} are kept, see the TAQ user guide for details.
Descriptive statistics are presented in Table 2 of the paper.

\setstretch{1.4}
%\bibliographystyle{chicago}
%\bibliography{how_to_data}



\end{document}

